Check for updates

OPEN ACCESS

EDITED BY Ralf Weiskirchen, RWTH Aachen University, Germany

REVIEWED BY

Yao Liu, Daping Hospital, China Peng Zhang, Beijing University of Chinese Medicine, China Arnab Mukherjee, Pfizer, United States

*CORRESPONDENCE Jinfang Zhang, is zhangjf06@gzucm.edu.cn Xixiao Yang, is yaxx@smu.edu.cn

RECEIVED 09 October 2024 ACCEPTED 10 December 2024 PUBLISHED 09 January 2025

CITATION

Su H, Xiao S, Liang Z, Xun T, Zhang J and Yang X (2025) Systematic review and bayesian network meta-analysis: comparative efficacy and safety of six commonly used biologic therapies for moderate-to-severe Crohn's disease. *Front. Pharmacol.* 15:1475222. doi: 10.3389/fphar.2024.1475222

COPYRIGHT

© 2025 Su, Xiao, Liang, Xun, Zhang and Yang. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Systematic review and bayesian network meta-analysis: comparative efficacy and safety of six commonly used biologic therapies for moderate-to-severe Crohn's disease

Haohang Su^{1,2}, Shengwei Xiao¹, Zhiqing Liang^{1,2}, Tianrong Xun¹, Jinfang Zhang^{3,4}* and Xixiao Yang^{1,5}*

¹Department of Pharmacy, Shenzhen Hospital, Southern Medical University, Shenzhen, China, ²School of Pharmaceutical Sciences, Southern Medical University, Guangzhou, China, ³Cancer Center, Shenzhen Hospital (Futian) of Guangzhou University of Chinese Medicine, Shenzhen, China, ⁴Shenzhen Traditional Chinese Medicine Oncology Medical Center, Shenzhen, China, ⁵Shenzhen Clinical Research Center for Digestive Disease, Shenzhen, China

Background: In contrast to previous network meta-analysis using classical frequentist methods, we evaluated the efficacy and safety of six frequently-used biologics through a Bayesian method.

Methods: Web of Science, Scopus, CENTRAL, ClinicalTrials.gov and ICTRP were searched to collect randomized controlled trials (RCTs) in adults with moderate-to-severe Crohn's disease, comparing Infliximab, Adalimumab, Certolizumab pegol, Ustekinumab, Risankizumab, or Vedolizumab, relative to placebo or an active comparator for induction of clinical response (two different definitions) and maintenance of clinical remission. A random-effects model was performed with rankings according to the surface under cumulative ranking curve (SUCRA) probability. Finally, we completed sensitivity and consistency analyses, and evaluated the certainty of evidence through GRADE working group guidance.

Results: We identified 22 and 20 RCTs for induction and maintenance therapy, respectively. Infliximab combined with azathioprine was most effective for inducing clinical response in TNF (tumor necrosis factor) antagonist-naïve patients. For TNF antagonist-experienced patients, Ustekinumab (SUCRA 86.19) and Risankizumab (SUCRA 62.56) have the largest SUCRA in induction of clinical response. Risankizumab has the lowest risk of adverse events (SUCRA

Abbreviations: ADA, Adalimumab; ADA + AZA, Adalimumab in combination with azathioprine; AEs, adverse events; CDAI, Crohn's disease activity index; CDAI-70/100, clinical response defined by a reduction in the Crohn's disease activity index ≥70/100 points compared to baseline; CD, Crohn's disease; CI, confidence interval; CrI, credible interval; CZP, Certollizumab pegol; IFX, Infliximab; IFX-5/10, the 5/10 mg/kg maintenance regimen of Infliximab; IFX + AZA, Infliximab in combination with azathioprine; MCMC, Markov chain Monte Carlo; PCO, placebo; RCTs, randomized controlled trials; RZB, Risankizumab; RZB-180/360, the 180/360 mg maintenance regimen of Risankizumab; SAEs, serious adverse events; SF, Supplementary File; SIs, serious infections; SUCRA, the surface under cumulative ranking curve; TNF, tumor necrosis factor; UST, Ustekinumab; VDZ, Vedolizumab; VDZ-SC, subcutaneous injections formulation of Vedolizumab.

84.81), serious adverse events (SUCRA 94.23), and serious infections (SUCRA 79.73) in induction therapy. Adalimumab and the 10 mg/kg regimen of Infliximab rank highest for maintaining clinical remission.

Conclusion: This analysis suggests that Infliximab in combination with azathioprine may be preferred biologic agents for induction therapy in TNF antagonist-naïve patients. For TNF antagonist-experienced patients, Ustekinumab and Risankizumab may be preferred biologic agents for induction therapy. Risankizumab potentially has the lowest safety risk worth exploring in induction therapy. Adalimumab and the 10 mg/kg regimen of Infliximab have maintenance efficacy benefits for responders to induction therapy.

Systematic Review Registration: https://www.crd.york.ac.uk/prospero/display_record.php?RecordID=458609, Identifier CRD42023458609.

KEYWORDS

bayesian network meta-analysis, Crohn's disease, biologic agents, efficacy, safety, GRADE

1 Introduction

Crohn's disease (CD) is a chronic inflammatory gastrointestinal disease characterized by multifactorial pathogenesis and refractoriness. Current therapies focus on reducing intestinal inflammation. As the discovery of novel drug targets and the increasing number of patients with CD refractory to conventional therapies, biologic therapies have gradually come to the forefront. However, with the introduction and clinical availability of infliximab (Knight et al., 1993; van Dullemen et al., 1995), the first tumor necrosis factor a (TNFa) antagonist for the treatment of CD, problems with its efficacy emerged. Among CD patients treated with TNFa antagonists, approximately two-thirds of the patients treated with TNFa antagonists are either primary non-response or loss of response (LOR) over time (Sandborn et al., 2012). Consequently, researchers have developed several new biologic agents based on different immune mechanisms to address the non-response to TNFa antagonists in CD patients: interleukin antagonists like Ustekinumab and Risankizumab (Benson et al., 2011; Singh et al., 2015), anti-integrin monoclonal antibody like Vedolizumab (Soler et al., 2009).

With the popularization of biologics with differing efficacy and safety profiles, clinicians have more therapeutic options when they make medical treatment plans for CD patients. However, they also face tough choices because of the uncertainty of the appropriate positioning of different agents in the disease course. In addition, it is a challenge for clinicians to determine the best pharmacotherapy for a patient who represents a primary non-response or secondary loss of response despite having undergone biologic therapy. These choices involve comprehensive decision-making analysis of the clinical rational medication incorporating patients' preferences as well as diagnosis and medical recommendations from clinicians. Therefore, researchers collect various clinical data through clinical trials and experiment with different treatment options to achieve clinical rational medication.

Nevertheless, today's redundant and variable-quality clinical trial results inevitably tire out the clinicians in need, there is a particular demand for systematic reviews to analyze valuable clinical trial results to provide a clear and concise clinical evidence base. Network Meta-Analysis (NMA), a statistical method for comparing multiple treatment options simultaneously through an integrated analysis of both direct and indirect evidence from RCTs, has provided a wealth of evidence on the safety and efficacy of therapies lacking head-to-head RCTs data for many clinical guidelines and Health Technology Assessments (HTAs).

In NMA, Bayesian methods offer several advantages over classical frequentist methods. First, Bayesian methods provide greater flexibility in handling uncertainty by incorporating prior beliefs through prior probability distributions and offering a more comprehensive representation of uncertainty in parameter estimation. Second, Bayesian methods offer advantages in dealing with heterogeneity, allowing for a more nuanced understanding of variability across studies and facilitating the use of complex models, thereby capturing data structure and patterns more effectively (Sutton and Abrams, 2001; Dias et al., 2014a; Higgins and Cochrane, 2019).

With new RCTs being initiated or completed on a daily basis, it is necessary to update the evidence collected, complied and presented by the NMA in order to provide the up-to-date evaluations. In this study, we conducted an updated systematic review and NMA with a Bayesian consistency DerSimonian-Laird random-effects model (Borenstein et al., 2010; Kanters, 2022; Watt and Del Giovane, 2022), using Markov chain Monte Carlo (MCMC) simulation, comparing the relative efficacy and safety of Infliximab, Adalimumab and Certolizumab pegol (three TNFa antagonists), Ustekinumab and Risankizumab (two interleukin antagonists), Vedolizumab (an anti-integrin monoclonal antibody), biosimilars (CT-P13 and BI 695501), and TNFa antagonists combined with immunosuppressants, for adult moderate-to-severe CD patients stratified by history of TNF antagonist treatment in the induction and maintenance therapy. We used Grading of Recommendations Assessment, Development, and Evaluation (GRADE) working group guidance to evaluate the certainty of evidence (Puhan et al., 2014).

2 Methods

The methods and recommendations of the PRISMA (Preferred Reporting Items for Systematic Reviews and Meta-Analysis)

extension statement for NMA for healthcare interventions, and the Cochrane Handbook were used to conduct this systematic review (Hutton et al., 2015; Higgins and Cochrane, 2019). Good research practices from the report of ISPOR (International Society for Pharmacoeconomics and Outcomes Research) for conducting indirect treatment comparisons and NMA for healthcare decision making were referenced (Hoaglin et al., 2011). The study protocol has been prospectively registered in PROSPERO (International Prospective Register of Systematic Reviews, www.crd.york.ac.uk/ prospero). The registration number is CRD42023458609.

2.1 Search strategy

A comprehensive search of multiple electronic databases, including PubMed, Embase, Web of Science, Scopus, CENTRAL (Cochrane Central Register of Controlled Trials), ClinicalTrials.gov and ICTRP (International Clinical Trials Registry Platform), from inception to 8 June 2023, without language or geographic area restrictions were conducted. MeSH/Emtree words combined with free words were used for the literature search (see Supplementary Table S1 in the Supplementary File 1 [SF1], full search strategy).

2.2 Selection criteria

Two investigators independently searched the study titles, abstracts and full texts of relevant studies, to identify articles of interest based on the following inclusion and exclusion criteria.

The selection criteria were in strict accordance with the PICOS (Patients, Intervention, Comparators, Outcomes, Study designs) principle. Study included in this NMA were phase II or III RCTs meeting the following criteria: (1) including adult (18-75 years of age) patients with moderate-to-severe CD (defined by CDAI 220-450); (2) receiving a biologic treatment with TNFa antagonists (Infliximab, Adalimumab or Certolizumab pegol), anti-interleukin (Ustekinumab, Risankizumab), anti-integrin (Vedolizumab), either alone or in combination with immunosuppressants, and their biosimilars, as the first-line biologic or after previous biologic exposure; (3) minimum duration of therapy of 4 weeks for trials reporting induction of response or remission in active disease and 22 weeks in trials reporting maintenance of response or remission; (4) compared to placebo or another medication for CD patients; (5) reporting induction of clinical remission (defined by a CDAI <150) or clinical response (defined by a reduction in the CDAI ≥70 points compared to baseline [CDAI-70], some studies defined the clinical response threshold of CDAI as 100 [CDAI-100], while some reported both CDAI-100 and CDAI-70), maintenance of remission or clinical response (among patients with clinical response to induction therapy in re-randomization trials and among all patients in treat-through trials), or safety (adverse events [AEs], serious adverse events [SAEs], and serious infections [SIs] as defined by the study authors). Only trials conducted in biologic and immunosuppressant naïve patients, in which patients were initiated on biologic monotherapy alone and compared with biologics and immunosuppressants in combination, were included as trials of combination therapy. In contrast, trials specifically enrolling immunosuppressant failed patients to evaluate a single biologic therapy or to compare two or more biologic therapies were included as trials of respective biologic monotherapy.

We excluded studies where: (1) trials of biologics to prevent postoperative recurrence; (2) trials comparing biologic treatment with surgical treatment; (3) pilot studies; (4) trials comparing therapeutic drug monitoring (TDM) with standard therapy; (5) trials with treatment failure or occurrence of complications as the primary end point; (6) trials of the effect of differences in baseline characteristics (C-reactive protein, prior anti-TNFa therapy, severity and disease distribution) on efficacy; (7) trials with unspecified interventions (a class but not a precise medication was specifical); (8) trials of the withdrawal of medications; (9) comparing regimen the optimized dosing bv pharmacokinetic dashboard (combined with TDM in the OPTIMIZE [Proactive infliximab optimization using а pharmacokinetic dashboard versus standard of care in patients with Crohn's disease] trial) with the standard dosing regimen; (10) trials of the effect of different preoperative medications on postoperative complications; (11) trials of dosing interval lengthening in maintenance phase; (12) early terminated trials; (13) trials without results; (14) trials included patients with other subtypes of inflammatory bowel diseases (IBD, CD is a subtypes of IBD); (15) reviews, letters, conference abstracts, animal studies; (16) trials of novel therapies without reported phase III data; (17) trials of advanced therapies not frequently used for the treatment of moderate-to-severe CD in clinical practice.

2.3 Data analysis

Two study investigators (Haohang Su and Shengwei Xiao) independently rated the risk of bias of included RCTs using RoB 2 (version 2 of the Cochrane risk-of-bias tool for randomized trials) (Sterne et al., 2019). Any discrepancies were resolved by consensus or in consultation with a third reviewer (Xixiao Yang). Five domains (randomization process, deviations from intended interventions, Missing outcome data, measurement of the outcome, selection of the reported result) were assessed separately as "low risk", "high risk", or "some concerns" and analyzed together to produce a final risk of bias judgement for each study (see Supplementary Figure S1 in SF1, traffic light plot of the risk of bias assessment). A sensitivity analysis was conducted excluding the inferior quality trials.

A standardized data extraction form was used to capture study-, participant-, disease-, and treatment-related characteristics, and was done independently by two investigators. Any discrepancies were resolved by consensus or in consultation with the third reviewer. The most complete report of trial data, based on intent-to-treat analysis principles for efficacy outcomes and last observation carried forward for safety outcomes, was used for extraction if study results were reported in multiple publications. In addition, Individual patientlevel data was not sought in this review.

The primary efficacy outcome chosen in induction trials included achievement of clinical response (defined by a reduction in the CDAI \geq 70 [CDAI-70] or \geq 100 [CDAI-100] points compared to baseline). Secondary outcome was achievement of clinical remission, defined by a CDAI <150. For maintenance trials, the primary efficacy outcome was maintenance of clinical remission,

secondary outcomes were CDAI-70 and CDAI-100. The results of evaluations of CDAI-100 as the predominant for interpretation when discrepancies between the results of CDAI-70 and CDAI-100 in the evaluations of clinical response occurred (Sands et al., 2024). Both responder re-randomization and treat-through trials were included, although data informing remission rates among induction responders was preferentially used when available for the primary maintenance outcome in treat-through studies. A sensitivity analysis was conducted excluding treat-through trials. The safety outcomes evaluated in induction or maintenance trials were adverse events (AEs), serious adverse events (SAEs), and serious infections (SIs), as defined by the primary study authors (included the timepoint for the safety summary).

Data from a medication's approved dosage and administration was used, but the included subcutaneous injections formulation of Vedolizumab (Vedolizumab SC) evaluated in the VISIBLE-2 trial is an exception because of its potential clinical advantage. In trials had multiple outcomes from different times, we extracted the data from week 6 for the induction phase (if data from week 6 were not available, the earliest data from week 4 onwards were extracted), and the latest result for the maintenance phase. Long-term open-label trials were not used. The denominator used in all trials was based on intent-to-treat analysis with dropouts or missing data treated as non-responders for remission and response outcomes; lastobservation-carried-forward (LOCF) or observed-case (OC) was used for missing continuous values; missing categorical variables were imputed using non-responder imputation (NRI).

Other extracted covariables included location of trial conduct, number of trials centers or sites, sample size (included the number of TNF antagonist-naïve patients and total sample size), trial design (received medication, dosage regimen, duration of intervention, and randomization), dosage and schedule of active/comparator group, severity of CD at randomization, definition of outcomes, outcome timepoints, concomitant medications (divided into active group and placebo/comparator group, medications included aminosalicylates, immunosuppressants, and corticosteroids).

A Bayesian random-effects model was built given the anticipated differences between trials with respect to patient enrolment criteria, patient populations, outcome timepoint of evaluations and interventions (assessments of model fit are presented in the results). A sensitivity analysis was conducted changing the original random-effects model to a Mantel-Haenszel fixed-effect model. Pooled odds ratios (ORs) with 95% credible interval (CrI) were estimated, and statistical heterogeneity was evaluated using the 1^2 statistic. We also conducted a sensitivity analysis using relative risk (RR) as the effect size.

In the Bayesian model for evaluating the induction and maintenance of clinical remission in moderate-to-severe CD patients, the MCMC parameters were set to 4 chains, 20,000 adaptation iterations, 200,000 total iterations, burn-in iterations were half of total iterations rounded down and the thin parameter was 1. The priors on the baseline and relative treatment effects were defined as independent normal priors (used with mean 0 and standard deviation 15u, where u is the largest maximum likelihood estimator in single trials), an uniform distribution with range 0 to u was used as the prior on variance of relative treatment effects (van Valkenhoef et al., 2012). In the other evaluations, the total iterations were set to 50,000. A sensitivity analysis was

conducted changing the original standard deviation of independent normal priors of the baseline treatment effects and relative treatment effects to 10u.

Statistical heterogeneity of these various outcomes was demonstrated by forest plots (see Supplementary Figures S2A-2E; Supplementary Figures S3-S7 in SF1, network plots and forest plots of direct comparisons). Small study effects and publication bias were examined by assessing for funnel plot asymmetry (see Supplementary Figures S8-S14 in SF1, all funnel plots). A sensitivity analysis was conducted excluding the trials which data points fell outside the 90% confidence interval in the funnel plots. Model fit was assessed by computing the Deviance Information Criteria (DIC) and producing a leverage plot, all leverage plots are presented in Supplementary Figures S15-S19 (Dias et al., 2014a). MCMC model convergence was assessed by Brooks-Gelman-Rubin diagnosis plot, computing shrink factor and potential scale reduction factor (PSRF), all assessment results of convergence are presented in Supplementary Figures S20-S24 in SF1. The relative ranking of agents for each outcome was expressed using rankogram (a kind of bar chart showing the probability of each treatment ranking in each rank) and the surface under the cumulative ranking curve (SUCRA). Higher SUCRA values correspond to better efficacy or safety outcomes for induction/maintenance (better means that the corresponding intervention is more effective or has lower safety risks). The relative effectiveness of biologics for each outcome was compared in league heat plot and forest plot. All plots of comparative evaluation results are presented in Supplementary Figures S25-S39 in SF1.

In inconsistency analysis, we plotted the posterior mean deviance contributions of the individual data points for the consistency model and the inconsistency model along with the line of equality (see Supplementary Figure S40 in SF1, all consistency vs. inconsistency plots) and analyzed the anomalous data points specifically (Dias et al., 2014b). Finally, we used GRADE working group guidance to evaluate the certainty of evidence (Puhan et al., 2014) (see Supplementary Tables S2–S6 in SF1). All data analysis described above were conducted using 'BUGSnet' package in R statistical software (version 4.3.1). (Beliveau et al., 2019).

3 Results

3.1 Search results

The search strategy yielded 16,252 citations, of which 3,214 were duplicates and removed. Of the remaining 13,038 records that were screened, 103 full text articles were reviewed 31 trials in 28 studies were eligible for inclusion (see Supplementary Figure S41 in SF1, flow diagram of selection). A total of 22 RCTs were involved in evaluations of induction efficacy and safety in patients with moderate-to-severe CD, of which 13 and 8 RCTs evaluated induction efficacy in TNF antagonist-naïve and -experienced patients, respectively. There were 20 RCTs evaluating maintenance efficacy and safety in patients with moderate-to-severe CD, including 11 and 6 RCTs evaluating maintenance efficacy in TNF antagonist-naïve and experienced patients, respectively.

Patient and trial-level characteristics for induction and maintenance are summarized in Supplementary Tables S7-S8 in

SF1, five of the included trials were completed in a single country, and the rest were completed in multiple centers or sites. Patient inclusion and exclusion criteria varied among these trials, mainly manifested in severity of CD (3 RCTs were defined as CDAI 220-400, the DIAMOND trial was defined as CDAI ≥220 (Matsumoto et al., 2016), all others were defined as CDAI 220-450), TNF antagonists using history (8 RCTs excluded TNF antagonist-experienced patients and 15 RCTs included patients with prior failure of TNF antagonists, the other trials required conventional therapy failure as an inclusion criterion), and concomitant medications. On the comparator settings, one head-to-head, active-comparator trial (SEAVUE) comparing Ustekinumab with Adalimumab was identified (Sands et al., 2022); two head-to-head, active-comparator trials (the SONIC, DIAMOND trials) comparing biologic with combination therapy (biologic combined with azathioprine) were identified (Colombel et al., 2010; Matsumoto et al., 2016); and two three-arm trials (the ACCENT-1 and FORTIFY trials) comparing different doses of Infliximab or Risankizumab were identified (Hanauer et al., 2002; Ferrante et al., 2022). One placebo-controlled trial evaluating Vedolizumab SC was included, although subcutaneous injections formulation has not been approved (by U.S. Food and Drug Administration [FDA]) as an administration of Vedolizumab in the treatment of CD. Two head-to-head, biosimilar-comparator trials (Infliximab and Adalimumab were each compared with their biosimilars [CT-P13 and BI 695501]) were identified (Ye et al., 2019; Hanauer et al., 2021). All of the included trials were re-randomization trials, except for the SONIC, DIAMOND, SEAVUE, PRECISE-1 trials, which were treat-through trials. All induction or maintenance outcomes were uniformly evaluated based on standard definition of the CDAI, between weeks 4 and 10 for induction therapy phase and weeks 22 and 60 for maintenance phase, and the results of secondary outcomes are presented in Supplementary Result 1-4 in SF1.

3.2 Efficacy in overall patients with Crohn's disease

3.2.1 Induction therapy

A total of 22 RCTs including 6,427 moderate-to-severe CD patients evaluated induction clinical response (CDAI-70 and CDAI-100) after treatment with Infliximab (n = 3 trials, one biosimilarcomparator trial [CT-P13] and one combination-therapycomparator trial [combined with azathioprine] were included) (Targan et al., 1997; Colombel et al., 2010; Ye et al., 2019), Adalimumab (n = 6 trials, one biosimilar-comparator trial [BI 695501] and one combination-therapy-comparator trial [combined with azathioprine] were included) (Hanauer et al., 2006; Sandborn et al., 2007c; Watanabe et al., 2012; Matsumoto et al., 2016; Chen et al., 2020; Hanauer et al., 2021), Certolizumab Pegol (n = 2 trials) (Sandborn et al., 2007a; Sandborn et al., 2011), Ustekinumab (n = 4 trials, included one active-comparator [Adalimumab] trial) (Sandborn et al., 2012; Feagan et al., 2016; Sands et al., 2022), Risankizumab (n = 3 trials) (Feagan et al., 2017; D'Haens et al., 2022), Vedolizumab (n = 4 trials) (Sandborn et al., 2013; Sands et al., 2014; Takeda, 2017; Watanabe et al., 2020).

In the evaluation of inducing CDAI-70 in moderate-to-severe CD patients, funnel plots did not show evidence of publication bias or

small study effects after including outcome timepoint as a moderator, and also suggests that this part of the heterogeneity caused by differences in outcome timepoint contributed to the asymmetry of the funnel plot. The Brooks-Gelman-Rubin diagnosis plot and the calculated PSRF indicated convergence of the MCMC sampling. On direct meta-analysis, all agents except Risankizumab and Vedolizumab, which CDAI-70 were not evaluated in included RCTs, were significantly superior to placebo for inducing CDAI-70. Neither Infliximab nor Adalimumab had significant superiority over their respective biosimilars or combinations with azathioprine. On network meta-analysis, Infliximab alone and Infliximab in combination with azathioprine were both significantly superior to Adalimumab, Ustekinumab, BI 695501, Adalimumab and azathioprine in combination, and Certolizumab Pegol. The evidence also supported a significant superiority of CT-P13 over Certolizumab Pegol (log OR 2.68 [95% CrI: 0.29, 5.21], Figure 1A). Overall, Infliximab in combination with azathioprine (SUCRA 95.21) and Infliximab alone (SUCRA 84.99) were ranked highest (Figure 1B).

In the evaluation of inducing CDAI-100 in moderate-to-severe CD patients, the results of funnel plot and model convergence assessments were within appropriate and controllable range. The leverage plot suggested that the SEAVUE trial may have contributed to the poor model fit and a sensitivity analysis was conducted excluding the SEAVUE trial. On direct meta-analysis, all agents except Infliximab (CDAI-100 were not evaluated in included RCTs) and Vedolizumab (OR 1.49 [95% CI: 0.94, 2.37]), were significantly superior to placebo for inducing CDAI-100. There was no benefit of Adalimumab over BI 695501 (OR 0.86 [95% CI: 0.35, 2.15]) for inducing CDAI-100, and similarly no benefit of Ustekinumab over Adalimumab (OR 1.09 [95% CI: 0.71, 1.67]). On network metaanalysis, Ustekinumab was significantly superior to Certolizumab Pegol (log OR 0.53 [95% CrI: 0.03, 1.03], Figure 1A). Ustekinumab (SUCRA 76.62), BI 695501 (SUCRA 75.47), and Adalimumab (SUCRA 71.66) were ranked highest (Figure 2B).

3.2.2 Maintenance therapy

Overall, 20 RCTs including 4,834 patients with moderate-tosevere CD evaluated maintenance clinical remission. Therapies that were evaluated included Infliximab (n = 4 trials, included one biosimilar-, one combination-therapy-comparator trial and one three-arm trial [the ACCENT-1 trial]) (Rutgeerts et al., 1999; Hanauer et al., 2002; Colombel et al., 2010; Ye et al., 2019), Adalimumab (n = 6 trials, included one biosimilar-comparator trial and one combination-therapy-comparator trial) (Colombel et al., 2007; Sandborn et al., 2007b; Rutgeerts et al., 2012; Watanabe et al., 2012; Matsumoto et al., 2016; Hanauer et al., 2021), Certolizumab Pegol (n = 2 trials) (Sandborn et al., 2007a; Schreiber et al., 2007), Ustekinumab (n = 3 trials, one activecomparator [Adalimumab] trial was included) (Sandborn et al., 2012; Feagan et al., 2016; Ferrante et al., 2022; Sands et al., 2022), Risankizumab (n = 1 trial, one three-arm trial [the FORTIFY trial]) (Ferrante et al., 2022), Vedolizumab (n = 4 trials, included one trial evaluating Vedolizumab SC) (Sandborn et al., 2013; Takeda, 2017; Watanabe et al., 2020; Vermeire et al., 2022).

In the evaluation of maintaining clinical remission in moderateto-severe CD patients, funnel plots did not show evidence of publication bias or small study effects, the Brooks-Gelman-Rubin

	0.45	-0.25	-2.05	-2.80	-2.78	-2.92	-2.43	N/A	N/A	-3.24
NI/A	(-0.72 to 1.63)	(-1.46 to 0.97) -0.69	(-4.08 to -0.16) -2.51	(-5.24 to -0.46) -3.25	(-5.39 to -0.27) -3.24	(-5.15 to -0.85) -3.37	(-4.49 to -0.57) -2.88	NVA	NUA	(-5.19 to -1.49) -3.69
IN/A	IFATAZA	(-2.38 to 0.99)	(-4.83 to -0.27) -1.82	(-5.95 to -0.63) -2.56	(-6.08 to -0.46) -2.54	(-5.90 to -1.00) -2.68	(-5.25 to -0.69) -2.19	N/A	IN/A	(-5.97 to -1.60) -3.00
N/A	N/A	CI-P13	(-4.16 to 0.45)	(-5.27 to 0.09)	(-5.41 to 0.25)	(-5.21 to -0.29)	(-4.57 to 0.04)	N/A	N/A	(-5.29 to -0.88)
N/A	N/A	N/A	ADA	(-2.15 to 0.63)	(-2.47 to 0.93)	(-2.20 to 0.37)	(-1.35 to 0.49)	N/A	N/A	(-1.88 to -0.60)
N/A	N/A	N/A	N/A	ADA+AZA	(-2.18 to 2.19)	(-2.04 to 1.73)	(-1.33 to 2.01)	N/A	N/A	(-2.00 to 1.06)
N/A	N/A	N/A	-0.16 (-1.19 to 0.86)	N/A	BI 695501	-0.14 (-2.26 to 1.96)	(-1.57 to 2.28)	N/A	N/A	-0.47 (-2.26 to 1.37)
N/A	N/A	N/A	0.48 (-0.04 to 1.06)	N/A	0.65 (-0.51 to 1.83)	CZP	0.48 (-0.82 to 1.80)	N/A	N/A	-0.33 (-1.45 to 0.79)
N/A	N/A	N/A	-0.04 (-0.44 to 0.39)	N/A	0.12 (-0.98 to 1.25)	-0.53 (-1.03 to -0.03)	UST	N/A	N/A	-0.81 (-1.48 to -0.15)
N/A	N/A	N/A	0.06 (-0.48 to 0.61)	N/A	0.22 (-0.94 to 1.40)	-0.42 (-1.00 to 0.12)	0.10 (-0.42 to 0.59)	RZB	N/A	N/A
N/A	N/A	N/A	0.39	N/A	0.55 (0.58 to 1.72)	-0.10	0.43	0.33	VDZ	N/A
N/A	N/A	N/A	0.83	N/A	1.00	0.34	0.87	0.77	0.44	PCO
			(0.47 to 1.23)		(-0.09 to 2.10)	(-0.04 to 0.73)	(0.56 to 1.19)	(0.39 to 1.18)	(0.08 to 0.78)	
							57.05			
	A	dalimumab -					37.03	71.66		
							60.86			
	Adalimumab+A	zathioprine -			28.48					
					27.61					
					29.74					
		BI 695501 -						75.4	47	
		BI 695501 -						68.04	47	
		BI 695501 -						68.04	47 78.74	
		BI 695501 - CT-P13 -						75.4 68.04	47 78.74 85.23	
		BI 695501 -		24.0	17			75.4 68.04	47 78.74 85.23	
	Certolizu	BI 695501 - CT-P13 - umab pegol -		24.0 24.0 24.0 24.0 24.0 24.0 24.0 24.0	17 5.71			68.04	47 78.74 85.23	
ions	Certolizu	BI 695501 - CT-P13 - umab pegol -		24.0 24 24 24 24 24 24 24 24 24 24 24 24 24	17 5.71			68.04	47 78.74 85.23 84.99	
ventions	Certolizu	BI 695501 - CT-P13 - umab pegol -		24.0 21 16.78	17 5.71			75. 68.04	47 78.74 85.23 84.99	8.62
Interventions	Certolizu	BI 695501 - CT-P13 - umab pegol -		24.0 21 16.78	7 5.71			75 68.04	47 78.74 85.23 84.99 84.99	8.62
Interventions	Certolizu Infliximab+A	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine -		24.0 22 16.78	7 5.71			68.04 75.	47 78.74 85.23 84.99 84.99	95.21
Interventions	Certolizu Infliximab+A	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine -		24.0 21 16.78	7 5.71			68.04	47 78.74 85.23 84.99 8	95.21 92.26
Interventions	Certolizu Infliximab+A	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine - Placebo -	9.64	24.0 21 16.78	17 5.71			68.04 75.	47 78.74 85.23 84.99 8	8.62 96.21 92.26
Interventions	Certolizu Infliximab+A	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine - Placebo -	9.64 1.42 2.91	24.0 21 16.78	17 5.71			75.4 68.04	47 78.74 85.23 84.99 84.99	8.62 95.21 92.26
Interventions	Certolizu Infliximab+A	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine - Placebo -	9.64 1.42 2.91	24.0 21 16.78	77 5.71			68.04 75.88	47 78.74 85.23 84.99	8.62 95.21 92.26
Interventions	Certolizu Infliximab+A	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine - Placebo -	9.84 1.42 2.91	24.0 22 16.78	77 5.71	8.71		65.88	47 78.74 85.23 84.99 8	8.62 95.21 92.26
Interventions	Certolizu Infliximab+A Ris	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine - Placebo -	9.64 2.91	24.0 22 16.78	77 5.71	8.71		65.88	47 78.74 85.23 84.99 8	8.62 95.21 92.26
Interventions	Certolizu Infliximab+A Ris Us	BI 695501 - CT-P13 - Jmab pegol - Infliximab - zathioprine - Placebo - ankizumab -	9.64 2.91	24.0 2 16.78	77 5.71	8.71 42.09 46.06		65.88 75.4	47 78.74 85.23 84.99 8	8.62 95.21 92.26
Interventions	Certolizu Infliximab+A Ris Us	BI 695501 - CT-P13 - Jmab pegol - Infliximab - zathioprine - Placebo - Jankizumab -	9.64 1.42 2.91	24.0 2 16.78	7 5.71	8.71 42.09 46.06		68.04 65.88	47 78.74 85.23 84.99 8	8.62 95.21 92.26
Interventions	Certolizu Infliximab+A Ris Us	BI 695501 - CT-P13 - umab pegol - Infliximab - zathioprine - Placebo - ankizumab - stekinumab -	9.64 1.42 2.91	24.0 16.78	33.25	8.71 42.09 46.06		68.04 65.88	47 78.74 85.23 84.99 8	8.62 96.21 92.26
Interventions	Certolizu Infliximab+A Ris Us Ve	BI 695501 - CT-P13 - Jinfliximab - Zathioprine - Placebo - ankizumab - stekinumab - edolizumab -	9.64 1.42 2.91	24.0 2 16.78 22.87	33.25	8.71 42.09 46.06		68.04 65.88 76	47 78.74 85.23 84.99 8 8	6.62 96.21 92.26
Interventions	Certolizu Infliximab+A Ris Us Ve	BI 695501 - CT-P13 - Jmab pegol - Infliximab - Zathioprine - Placebo - ankizumab - stekinumab - edolizumab -	9.64 1.42 2.91	24.0 22 16.78 22.87 20	77 5.71 33.25	8.71 42.09 46.06	é0	68.04 65.88 76	47 78.74 85.23 94.99 8 8 8 8	8.62 95.21 92.26

(A) SUCRA (the surface under cumulative ranking curve) values of each intervention in efficacy evaluations of induction therapy in overall patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most effective. Conversely, the closer the value is to 0, the less effective the treatment is likely to be. (B) League heat plot of mixed comparisons of inducting clinical response (defined by a reduction in the CDAI [Crohn's Disease Activity Index] \ge 70 [CDAI-70] or \ge 100 [CDAI-100] points compared to baseline) between included biologic agents in overall patients with moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or less than 0 are indicated in red/yellow or blue/ green, respectively, in comparisons of inducing CDAI-100/CDAI-70, and the darker the color, the greater the odds ratios. Bold type represents statistically significant superiority/inferiority of the intervention over the comparator.

diagnosis plot and the calculated PSRF indicated convergence of the MCMC sampling. On direct meta-analysis, all agents included Vedolizumab SC (OR 1.77 [95% CI: 1.15, 2.71]) were significantly superior to placebo. Comparisons between treatment groups of biologic agents (5 mg/kg maintenance regimen of Infliximab [IFX-5] vs. 10 mg/kg maintenance regimen of Infliximab [IFX-5] vs. 10 mg/kg maintenance regimen of Risankizumab [RZB-180] vs. 360 mg maintenance regimen of Risankizumab [RZB-360] in the FORTIFY trial) were not statistically different in either of

the two included three-arm trials. Neither the corresponding biosimilar nor combination of azathioprine had a significant effect on the efficacy of infliximab or adalimumab compared with monotherapy. There was no benefit of Ustekinumab over Adalimumab (OR 1.18 [95% CI: 0.78, 1.79]). On network metaanalysis, Adalimumab (SUCRA 81.05), IFX-10 (SUCRA 76.60), Infliximab in combination with azathioprine (SUCRA 72.81), Adalimumab in combination with azathioprine (SUCRA 64.98) were ranked highest (Figure 2).



(A) SUCRA (the surface under cumulative ranking curve) values of each intervention in efficacy evaluation of maintenance therapy in overall patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most effective. Conversely, the closer the value is to 0, the less effective the treatment is likely to be. (B) League heat plots of mixed comparisons of maintaining clinical remission (defined by a CDAI [Crohn's Disease Activity Index] < 150) between included biologic agents in overall patients with moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or less than 0 are indicated in red or blue, respectively, in comparisons of maintaining clinical remission, and the darker the color, the greater the odds ratios. Bold type represents statistically significant superiority/inferiority of the intervention over the comparator.

٨								
IFX	0.45	-0.24	-1.63	-2.39	N/A	N/A	N/A	-3.23
N/A	(-2.54 to 3.42) IFX+AZA	(-3.27 to 2.75) -0.69 (-4.93 to 3.54)	(-5.19 to 2.22) -2.09 (-6.65 to 2.84)	(-7.01 to 2.56) -2.84 (-8.32 to 2.96)	N/A	N/A	N/A	(-6.49 to -0.05) -3.68 (-8.09 to 0.69)
N/A	N/A	CT-P13	-1.40 (-5.97 to 3.55)	-2.15 (-7.61 to 3.71)	N/A	N/A	N/A	-3.00 (-7.38 to 1.39)
N/A	N/A	N/A	ADA	-0.75 (-3.82 to 2.28)	N/A	N/A	N/A	-1.59 (-3.60 to 0.01)
N/A	N/A	N/A	N/A	ADA+AZA	N/A	N/A	N/A	-0.85 (-4.58 to 2.50)
N/A	N/A	N/A	0.65 (-0.11 to 1.49)	N/A	CZP	N/A	N/A	N/A
N/A	N/A	N/A	-0.02 (-0.60 to 0.59)	N/A	-0.67 (-1.49 to 0.09)	UST	N/A	N/A
N/A	N/A	N/A	0.46 (-0.40 to 1.34)	N/A	-0.20 (-0.99 to 0.56)	0.48 (-0.38 to 1.33)	VDZ	N/A
N/A	N/A	N/A	1.03 (0.43 to 1.69)	N/A	0.38 (-0.12 to 0.87)	1.05 (0.46 to 1.68)	0.57 (-0.01 to 1.19)	PCO

B.



FIGURE 3

(A) SUCRA (the surface under cumulative ranking curve) values of each intervention in efficacy evaluations of induction therapy in tumor necrosis factor antagonist-naïve patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most effective. Conversely, the closer the value is to 0, the less effective the treatment is likely to be. (B) League heat plots of mixed comparisons of inducing clinical response (defined by a reduction in the CDAI [Crohn's Disease Activity Index] \geq 70 [CDAI-70] or \geq 100 [CDAI-100] points compared to baseline) between included biologic agents in tumor necrosis factor antagonist-naïve patients with moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or lost than 0 are indicated in red/yellow or blue/green, respectively, in comparisons of inducing CDAI-100/CDAI-70, and the darker the color, the greater the odds ratios. Bold type represents statistically significant superiority/inferiority of the intervention over the comparator.

3.3 Efficacy in TNF antagonist-naïve patients with Crohn's disease

3.3.1 Induction therapy

A total of 14 induction RCTs including 3,050 TNF antagonistnaïve patients with moderate-to-severe CD evaluated induction clinical response (CDAI-70 and CDAI-70) after treatment with Infliximab (n = 3 trials, included one biosimilar-, one combination-therapy-comparator trial) (Targan et al., 1997; Colombel et al., 2010; Ye et al., 2019), Adalimumab (n = 4 trials, included one combination-therapy-comparator trial) (Hanauer et al., 2006; Watanabe et al., 2012; Matsumoto et al.,

2016; Chen et al., 2020), Certolizumab Pegol (n = 2 trials) (Sandborn et al., 2007a; Sandborn et al., 2011), Ustekinumab (n = 2 trials, included one active-comparator [Adalimumab] trial) (Feagan et al., 2016; Sands et al., 2022), Vedolizumab (n = 3 trials) (Sandborn et al., 2013; Sands et al., 2014; Watanabe et al., 2020).

In the evaluation of inducing CDAI-70 in TNF antagonist-naïve CD patients, the results of funnel plot, model fit and convergence assessments do not require any special attention. On direct metaanalysis, Infliximab (OR 22.00 [95% CI: 5.17, 93.56]) and Adalimumab (OR 4.46 [95% CI: 2.08, 9.56]) were significantly superior to placebo, and the other biologic agents were not evaluated for CDAI-70 in the included RCTs. Similarly, for both Infliximab (OR 0.64 [95% CI: 0.40, 1.02]) and Adalimumab (OR 2.05 [95% CI: 0.83, 5.08]), there was no benefit of their monotherapies over combinations with azathioprine, and Infliximab did not differ significantly from CT-P13 (OR 1.28 [95% CI: 0.71, 2.30]) for inducing CDAI-70. On network meta-analysis, Infliximab in combination with azathioprine (SUCRA 82.68) and Infliximab alone (SUCRA 72.54) were ranked highest (Figure 3).

In the evaluation of inducing CDAI-100 in TNF antagonistnaïve CD patients, all but Infliximab, which CDAI-100 was not evaluated in included RCTs, were significantly superior to placebo on direct meta-analysis, and there was no benefit of Ustekinumab over Adalimumab (OR 1.09 [95% CI: 0.71, 1.67]). On network meta-analysis, Ustekinumab (SUCRA 84.50) and Adalimumab (SUCRA 82.08) were ranked highest (Figure 3).

3.3.2 Maintenance therapy

Only maintenance clinical remission was evaluated for TNF antagonist-naïve/experienced patients due to the lack of clinical data on the maintenance CDAI-70 and CDAI-100 to form a network. Overall, 11 RCTs including 2,335 TNF antagonist-naïve patients with moderate-to-severe CD evaluated maintenance clinical remission. Therapies that were evaluated included Infliximab (n = 4 trials, included one biosimilar-comparator trial, onecombination-therapy-comparator trial and on three-arm trial [the ACCENT-1 trial]) (Rutgeerts et al., 1999; Hanauer et al., 2002; Colombel et al., 2010; Ye et al., 2019), Adalimumab (n = 2 trials, included one combination-therapy-comparator trial) (Colombel et al., 2007; Matsumoto et al., 2016), Ustekinumab (n = 2 trials, included one active-comparator trial) (Feagan et al., 2016; Sands et al., 2022), Vedolizumab (n = 3 trials, included one trial evaluating Vedolizumab SC) (Sandborn et al., 2013; Watanabe et al., 2020; Vermeire et al., 2022).

In the evaluation of maintaining clinical remission in TNF antagonist-naïve patients, the results of funnel plot, model fit and convergence assessments do not require any special attention. On direct meta-analysis, all but Ustekinumab (OR 1.96 [95% CI: 0.89, 4.34]) and Vedolizumab SC (OR 1.26 [95% CI: 0.67, 2.36]) were significantly superior to placebo, and comparisons between active agent regimens did not show significant differences. On network meta-analysis, IFX-10 (SUCRA 68.98), Infliximab in combination with azathioprine (SUCRA 67.01), and Adalimumab alone (SUCRA 66.56) were ranked highest (Figures 4A, B).

3.4 Efficacy in TNF antagonist-experienced patients with Crohn's disease

3.4.1 Induction therapy

Overall, 10 RCTs including 2,131 TNF antagonist-experienced patients with moderate-to-severe CD evaluated induction clinical response (CDAI-70 and CDAI-70). Therapies that were evaluated included Adalimumab (n = 2 trials) (Sandborn et al., 2007c; Watanabe et al., 2012), Certolizumab Pegol (n = 1 trial) (Sandborn et al., 2007a), Ustekinumab (n = 3 trials) (Sandborn et al., 2012; Feagan et al., 2016), Risankizumab (n = 1 trial) (Feagan et al., 2017), and Vedolizumab (n = 3 trials) (Sandborn et al., 2013; Sands et al., 2014; Watanabe et al., 2020).

In the evaluation of inducing CDAI-70 in TNF antagonistexperienced CD patients, the results of funnel plot, model fit and convergence assessments do not require any special attention. On direct meta-analysis, only two placebo-controlled trials each of Adalimumab (OR 2.09 [95% CI: 1.34, 3.27]) and Ustekinumab (OR 1.79 [95% CI: 1.23, 2.58]) were included, and both biologics had a significantly superiority over placebo. Direct and network comparisons for induction CDAI-70 in TNF antagonist-experienced CD patients were similar, and the efficacy of Adalimumab was higher than that of Ustekinumab (Figure 5).

In the evaluation of inducing CDAI-100 in TNF antagonistexperienced CD patients, all but Risankizumab (OR 2.55 [95% CI: 0.86, 7.60]) and Certolizumab Pegol (OR 1.30 [95% CI: 0.64, 2.62]) were significantly superior to placebo on direct meta-analysis. On network meta-analysis, Ustekinumab (SUCRA 86.19) and Risankizumab (SUCRA 62.56) were ranked highest (Figure 5).

3.4.2 Maintenance therapy

Only maintenance clinical remission was evaluated for TNF antagonist-naïve/experienced patients. A total of 5 RCTs including 866 TNF antagonist-experienced patients with moderate-to-severe CD evaluated maintenance clinical remission after treatment with Adalimumab (n = 1 trial), Ustekinumab (n = 2 trials), Vedolizumab (n = 2 trials, included one trial evaluating Vedolizumab SC).

In the evaluation of maintaining clinical remission in TNF antagonist-experienced patients, all agents were significantly superior to placebo on direct meta-analysis. On network meta-analysis, Adalimumab (SUCRA 80.09) and Vedolizumab (SUCRA 60.25) were ranked highest (Figure 4C).

3.5 Safety in overall patients with Crohn's disease

Due to the lack of safety subgroup analysis data on the medication history of TNF antagonist, only the safety of each biologic agents in overall CD patients was evaluated.

3.5.1 Induction therapy

In the safety evaluation of patient with moderate-to-severe CD in induction therapy, direct and network comparisons for risk of AEs, SAEs, or SIs in induction therapy were similar, with no statistical differences between any of the included interventions. On network meta-analysis, Risankizumab (SUCRA 84.81), Adalimumab (SUCRA 71.67), and Ustekinumab (SUCRA 52.64)



(A) SUCRA (the surface under cumulative ranking curve) values of each intervention in evaluation of maintaining clinical remission in tumor necrosis factor antagonist-naïve/experienced patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most effective. Conversely, the closer the value is to 0, the less effective the treatment is likely to be. (B) League heat plots of mixed comparisons of maintaining clinical remission (defined by a CDAI [Crohn's Disease Activity Index] < 150) between included biologic agents in tumor necrosis factor antagonist-naïve patients with moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or less than 0 are indicated in red or blue, respectively, in comparisons of maintaining clinical remission of maintaining clinical remission (defined by a CDAI [Crohn's Disease Activity index] < 150) between included biologic agents in tumor necrosis factor antagonist-naïve patients with moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or less than 0 are indicated in red or blue, respectively, in comparisons of maintaining clinical remission, and the darker the color, the greater the odds ratios. Bold type represents statistically significant superiority/inferiority of the intervention over the comparator. (C) League heat plots of mixed comparisons of maintaining clinical remission (defined by a CDAI [Crohn's Disease Activity Index] < 150) between included biologic agents in tumor necrosis factor antagonist-experienced patients with moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or less than 0 are (*Continued*)

FIGURE 4 (Continued)

indicated in red or blue, respectively, in comparisons of maintaining clinical remission, and the darker the color, the greater the odds ratios. Bold type represents statistically significant superiority/inferiority of the intervention over the comparator.

had the lowest risk of AEs during induction therapy. Risankizumab (SUCRA 94.23), Adalimumab (SUCRA 74.72), and Ustekinumab (SUCRA 49.49) had the lowest risk of SAEs during induction therapy, and Risankizumab had a significant safety benefit over Ustekinumab, Vedolizumab, and Certolizumab Pegol. Risankizumab (SUCRA 79.73), Certolizumab Pegol (SUCRA 62.88), and Adalimumab (SUCRA 59.03) had the lowest risk of SIs during induction therapy (Figure 6A).

3.5.2 Maintenance therapy

In the safety evaluation of patients with moderate-to-severe CD in maintenance therapy, there was no significantly difference between included interventions in evaluation of the risk of AEs on direct meta-analysis. In the SONIC trial (Infliximab vs. Infliximab combined with azathioprine), Infliximab alone (5 mg/kg maintenance regimen of Infliximab) had a significantly higher risk of SAEs than in combination with azathioprine (OR 1.77 [95% CI: 1.03, 3.05]), while the other comparisons between the agents were not significantly different. On network meta-analysis, IFX-10 (SUCRA 73.71), Vedolizumab SC (SUCRA 59.74), and RZB-180 (SUCRA 56.31) had the lowest risk of AEs during maintenance therapy. BI 695501 (SUCRA 97.58), Ustekinumab (SUCRA 71.48), Infliximab in combination with azathioprine (SUCRA 70.48) had the lowest risk of SAEs during maintenance therapy, and BI 695501 had a significant safety superiority over almost all of the other agents (except Ustekinumab and Infliximab in combination with azathioprine). Vedolizumab SC (SUCRA 84.59), RZB-180 (SUCRA 59.46), and IFX-10 (SUCRA 58.85) had the lowest risk of SIs during maintenance therapy (Figure 6B), and Vedolizumab SC had a significant lower risk of SIs compared to Certolizumab Pegol (log OR -2.22 [95% CrI: -4.42, -0.17]).

3.6 Sensitivity analysis and consistency analysis

We conducted several sensitivity analyses and the general consistency of the results confirms the robustness of our findings (all sensitivity analyses results are detailed in Supplementary File 2). In consistency vs. inconsistency plots, the contributions to the deviance are very similar and close to 1, indicating no evidence of inconsistency in our networks.

4 Discussion

In this updated systematic review and Bayesian network metaanalysis, we combined direct and indirect evidence from 31 RCTs to evaluate the efficacy of different therapies for inducing and maintaining clinical remission or clinical response in moderateto-severe CD populations. Several important findings from this

network meta-analysis may inform clinical decisions. Firstly, two TNF antagonists, Infliximab and Adalimumab, continue to have a high status in induction and maintenance therapy. In addition, Infliximab in combination with azathioprine has a significant superiority in induction therapy in TNF antagonist-naïve patients, and appropriately increasing the dose of infliximab (10 mg/kg) may improve the efficacy of maintenance therapy. Secondly, Ustekinumab and Risankizumab may be the therapeutic agents of choice for TNF antagonist-experienced patients. High-dose Risankizumab (360 mg) also failed to show more efficacy benefits than low-dose Risankizumab (180 mg) in maintenance therapy. Thirdly, Vedolizumab does not offer superior therapeutic safety commensurate with its mechanism, and the subcutaneous injection formulation of Vedolizumab needs to demonstrate advantages beyond efficacy to establish its superiority over the widely approved intravenous infusion formulation. Finally, Risankizumab may be have worth-exploring lower safety risks in induction therapy. Altogether, these findings will inform the optimal sequencing of biologic therapy for patients with different histories of biologic administration.

Despite the emergence of new biologic agents, TNF antagonists remain most extensively experienced in the therapy of CD. In addition to the high-ranking performance in this network metaanalysis, TNF antagonists play a defined role in patients with fistulizing or structuring CD (Present et al., 1999; Sands et al., 2004; Bouhnik et al., 2018; Lee et al., 2018; Rodríguez-Lago et al., 2020), and seems to be a valuable option in the treatment of extraintestinal manifestations given the good response rates in a retrospective analysis of the Swiss IBD Cohort Study (Vavricka et al., 2017), though we acknowledge estimates supporting a large effect size for Infliximab have a low confidence due to the small number of induction RCTs available for inclusion and the small sample size of RCTs that were included. There was no statistically significant difference between 5 and 10 mg/kg intravenous infusion of Infliximab, which are both approved for the treatment of CD. However, in the three-arm ACCENT-1 trial, high-dose Infliximab (IFX-10) was associated with a superiority of SUCRA compared to the low-dose (IFX-5) in maintenance therapy (Hanauer et al., 2002). Combined with the fact that there is currently no evidence of higher safety risks associated with higher doses of Infliximab, this seems to suggest that increasing the dose of Infliximab has the potential to improve efficacy in maintenance therapy. While also as a TNF antagonist, Certolizumab Pegol did not show similar efficacy in the treatment of CD in this NMA. Nevertheless, in a retrospective analysis compared the real-world efficacy and safety of different TNF antagonists in biologic-naïve patients with CD, Certolizumab Pegol was comparable in efficacy to Adalimumab (Singh et al., 2016), and a head-to-head active-comparator RCT is required to judge the precise efficacy of Certolizumab Pegol.

Though the irreplaceable role of TNF antagonists in the treatment of CD, unfortunately one-third of patients with CD do

Α.					
ADA	N/A	-0.11 (-1.21 to 0.91)	N/A	N/A	-0.79 (-1.65 to -0.02)
0.58 (-1.50 to 2.84)	CZP	N/A	N/A	N/A	N/A
-0.61 (-2.12 to 1.15)	-1.18 (-3.13 to 0.81)	UST	N/A	N/A	-0.67 (-1.38 to 0.00)
-0.12 (-2.39 to 2.24)	-0.71 (-3.30 to 1.83)	0.48 (-1.71 to 2.56)	RZB	N/A	N/A
0.27 (-1.29 to 2.02)	-0.31 (-2.29 to 1.67)	0.87 (-0.53 to 2.23)	0.40 (-1.73 to 2.59)	VDZ	N/A
0.84 (-0.38 to 2.26)	0.27 (-1.45 to 1.97)	1.45 (0.47 to 2.39)	0.97 (-0.91 to 2.91)	0.57 (-0.42 to 1.57)	РСО





(A) SUCRA (the surface under cumulative ranking curve) values of each intervention in efficacy evaluations of induction therapy in tumor necrosis factor antagonist-experienced patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most effective. Conversely, the closer the value is to 0, the less effective the treatment is likely to be. (B) League heat plots of mixed comparisons of inducing clinical response (defined by a reduction in the CDAI [Crohn's Disease Activity Index] \geq 70 [CDAI-70] or \geq 100 [CDAI-100] points compared to baseline) between included biologic agents in tumor necrosis factor antagonist-experienced moderate-to-severe Crohn's disease, reporting odds ratios on the logarithmic scale (log OR) with 95% confidence intervals. Log OR greater or less than 0 are indicated in red/yellow or blue/green, respective/y, in comparisons of inducing CDAI-100/CDAI-70, and the darker the color, the greater the odds ratios. Bold type represents statistically significant superiority/inferiority of the intervention over the comparator.

not respond to initial treatment with TNF antagonists, and another one-third have only a transient response requiring a dose increase or switching to another therapy (Hanauer et al., 2002; Hanauer et al., 2006; Colombel et al., 2007; Sandborn et al., 2007a; Schreiber et al., 2007). Preventative combination treatment comes into view and may be the management of non-response. In our analysis,



(A) SUCRA (the surface under cumulative ranking curve) values of each intervention in safety evaluations of induction therapy in overall patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most safety. Conversely, the closer the value is to 0, the less effective the treatment is likely to be. (B) SUCRA (the surface under cumulative ranking curve) values of each intervention in safety evaluations of maintenance therapy in overall patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention in safety evaluations of maintenance therapy in overall patients with moderate-to-severe Crohn's disease. SUCRA values are expressed on a scale from 0 to 100. The closer the value is to 100, the more likely the intervention is to be among the most safety. Conversely, the closer the value is to 0, the less effective the treatment is likely to be.

efficacy benefit whereas Adalimumab in combination with azathioprine did not. In the SONIC trial, TNF antagonist-naïve patients with moderate-to-severe CD who received Infliximab in combination with azathioprine had a significant benefit on the rates of corticosteroid-free clinical remission compared to those receiving Infliximab alone or azathioprine alone, with comparable toxicity among groups (Colombel et al., 2010). In contrast, the open-label DIAMOND trial in which the efficacy of Adalimumab in combination with azathioprine did not differ significantly from that of Adalimumab monotherapy in patients naïve to both TNF antagonists and immunosuppression (Matsumoto et al., 2016). In both trials, patients receiving combination therapy who experienced worsening of CD were significantly fewer than those receiving TNF antagonist monotherapy (p = 0.045 in SONIC, p< 0.001 in DIAMOND, using Fisher's Exact Test). This suggests that combination therapy may help prevent LOR to TNF antagonists, although LOR is not the only cause of worsening of CD (Ben-Horin and Chowers, 2011).

However, the populations of DIAMOND and SONIC were naïve to immunosuppressants and biologics, meaning that the effect of combination therapy in preventing LOR in immunosuppressantand biologic-experienced patients remains unknown. Therefore, some researchers evaluated the benefit to risk ratio of concomitant immunosuppressives with scheduled TNF antagonists by identifying the influence of withdrawal of immunosuppressives. In an open-label interventional study reported by Van Assche G et al., there was no significant difference to continuation of immunosuppression in combination for < or = 6 months compared to scheduled Infliximab monotherapy in the proportions of patients with secondary non-response (Van Assche et al., 2008). Continuation of Adalimumab combined with oral thiopurines beyond 6 months offers no clear benefit over scheduled Adalimumab monotherapy in the DIAMOND-2 trial (Hisamatsu et al., 2019). Similarly, there is no evidence of superior efficacy of concomitant immunomodulating medication in therapy with non-TNF antagonistic biologics but come with an increased risk of SIs (Hu et al., 2021; Yzet et al., 2021). Overall, the role of concomitant immunosuppressives in treatment of CD still needs to be explored further.

In recent years, patents of Infliximab and Adalimumab have expired in many countries, and the huge biologics market has made biosimilar manufacturers eager. Both biosimilars, CT-P13 and BI 695501, were comparable in efficacy and safety to their respective originators in direct head-to-head comparisons (Ye et al., 2019; Hanauer et al., 2021). Subcutaneous injections formulation of CT-P13 (CT-P13 SC) is a product worthy of attention. The pharmacokinetics, as manifested by trough concentrations of the agents, increased after CT-P13 switching from intravenous to subcutaneous administration in a phase 1 study (Schreiber et al., 2021). Smith PJ et al. observed high treatment persistence rates and low rates of immunogenicity in IBD patients with maintenance therapy switching from Infliximab to CT-P13 SC in a multicenter retrospective cohort study (Smith et al., 2022). The results of the LIBERTY-CD trial demonstrate that maintenance therapy with CT-P13 SC provides both a robust clinical benefit and the convenience of subcutaneous administration to moderate-to-severe CD patients (Colombel et al., 2023).

In this network meta-analysis, although Adalimumab had the largest SUCRA value in the evaluations of inducing clinical remission and CDAI-70 in TNF antagonist-experienced CD patients, we recommend Ustekinumab and Risankizumab as the therapeutic agents of choice for induction therapy in TNF antagonist-experienced patients, because of the intransitivity caused by the inclusion criteria, which selectively included the patients with intolerance or secondary non-response to Infliximab, in the GAIN trial (Sandborn et al., 2007c). In contrast, the SEAVUE trial, a head-to-head trial to standard doses of different biologic agents for CD treatment, strengthens the reliability of our findings (Sands et al., 2022). The results of inconsistency analysis showed that there was no inconsistency between direct and indirect evidence of the comparison of Adalimumab and Ustekinumab (Dias et al., 2010; Dias et al., 2014b).

Notably, the high clinical benefit in SEAVUE compared to other RCTs may be attributed to several factors: the short disease course, the biologic-naïve baseline characteristics of enrolled patients (received early biologic treatment [EBT]), and non-placebo-controlled design (patients were aware they received active study therapy) (Sands et al., 2022). But in fact, the superiority of EBT over conventional therapy (CT) for CD is disputed. The result of an upgrade of a previous CADTH (Canadian Agency for Drugs and Technologies in Health) rapid response reported by Thompson W et al. suggested that the clinical effectiveness of EBT compared to CT for CD in adults is still unclear but did point to possible benefits which require further study (Thompson and Argaez, 2019). In contrast, Ungaro RC et al. demonstrated that EBT was associated with improved clinical outcomes compared to late or CT in both prospective clinical trials and real-world studies in a systematic review (Ungaro et al., 2020). In addition, early TNF antagonist therapy is a more cost-effective management of CD compared with conventional therapy (Marchetti et al., 2013; Beilman et al., 2019). However, it is worth noting that the results of these studies may not be generalizable to other biologics as they mainly involved EBT of TNF antagonists.

Between Ustekinumab and Risankizumab, in the SEQUENCE trial (Nct, 2020), a phase 3 head-to-head study compared these two interleukin antagonists for the treatment of patients with moderateto-severe CD who have failed at least one TNF antagonists. Risankizumab had demonstrated non-inferiority for clinical remission at week 24 and superiority of endoscopic remission at week 48 over Ustekinumab, and showed superiority versus Ustekinumab for all ranked secondary endpoints, including achievement of clinical remission at week 48, endoscopic response at week 48 and 24, steroid-free endoscopic remission at week 48, and steroid-free clinical remission at week 48, with no new safety risks identified. These results are posted on AbbVie News Center. Ustekinumab and Risankizumab actually differ in mechanisms of action, although they are both interleukin antagonists. Unlike Ustekinumab, which inhibits the biological activity of interleukin-12 (IL-12) and interleukin-23 (IL-23) by blocking the common p40 subunit of these two cytokines (Benson et al., 2011), Risankizumab downregulates IL-23mediated inflammation by binding with high affinity to the p19 subunit of IL-23 (Singh et al., 2015). Therefore, it does not affect the IL-12-dependent T-cell pathways that are implicated in infection and cancer risk (Brunda et al., 1993; Stobie et al., 2000), potentially resulting in a lower safety risk (Wong and Cross, 2019).

Sadly, detailed data from the SEQUENCE trial is not available, which may have resulted in the potential efficacy superiority of Risankizumab over Ustekinumab not being reflected in our analysis. Nevertheless, we obtained and analyzed the key information from several included placebo-controlled trials studying Risankizumab and come up with the following findings (Feagan et al., 2017; D'Haens et al., 2022; Ferrante et al., 2022): First, Risankizumab has a ranking superiority over other biologics on induction of clinical remission in TNF antagonist-experienced CD patients, which may be paralleled by the result of the newly completed SEQUENCE trial; Second, the efficacy of high dose of Risankizumab (360 mg) was no better than that of low dose (180 mg) in the maintenance therapy; Third, Risankizumab has a relatively low safety risk in induction therapy, which may be associated with the specific mechanism of action.

Compared to the other biologics, Vedolizumab provides a more acceptable safety profile in its mechanism. By specifically targeting the a4β7/MAdCAM-1 pathway, Vedolizumab reduces localized immune responses in the gut, thereby providing therapeutic effects against intestinal inflammation. This mechanism allows for treatment efficacy in the gut without interfering with T-cell trafficking to the central nervous system or compromising systemic immune function (Wyant et al., 2016). However, Vedolizumab did not stand out either on efficacy or safety in this analysis. In other words, there is no superiority of Vedolizumab over interleukin antagonists for induction therapy in TNF antagonist-experienced patients. Although we acknowledge that a head-to-head comparison would be required to firmly confirm this conclusion, the results of several recent real-world studies have also demonstrated that Ustekinumab has a better efficacy in CD patients who have failed treatment with TNF antagonists compared to Vedolizumab (Alric et al., 2020; Biemans et al., 2020; Townsend et al., 2020; Manlay et al., 2021; Garcia et al., 2024). For example, García MJ et al. reported a prospective study including 207 Vedolizumab-treated patients and 628 Ustekinumab-treated patients who had TNF antagonist failure or intolerance from ENEIDA registry (Nationwide study on genetic and environmental determinants of inflammatory bowel disease): Vedolizumab was associated with a higher risk of treatment discontinuation compared with Ustekinumab, adjusted by corticosteroids at baseline, moderate-severe activity in Harvey-Bradshaw Index (HBI), and high levels of C-reactive protein at baseline (Garcia et al., 2024). The potential differences on efficacy between Ustekinumab and Vedolizumab in patients with prior failure of TNF antagonists may be attributed to different mechanisms for down-regulating IL-23-driven compensatory inflammation after blockade of TNF pathway (Schmitt et al., 2019). Vedolizumab SC may have superiority on safety and patient medication compliance because of its method of administration and was approved for maintenance therapy of CD in several countries, although it did not have benefit on efficacy in our network meta-analysis. Nevertheless, in the future, the management of switching the intravenous infusions formulation to the subcutaneous injections formulation in maintenance therapy phase deserves attentions.

In this meta-analysis, we also collected and analyzed the safety outcomes of the included RCTs and the results showed the superiority of Risankizumab and BI 695501 in induction and maintenance therapy, respectively. However, these findings should be interpreted cautiously for several reasons. First, these safety findings are at odds with the results of some recent real-world studies. For example, the results of multicenter California-IBD cohort study reported by Singh S et al. suggested that Ustekinumab was associated with a lower risk of SIs in CD patients compared with TNF antagonists and Vedolizumab, but the risk of SIs in maintenance therapy with Ustekinumab, Vedolizumab and three original TNF antagonists did not significantly different in our analysis (Singh et al., 2023). Vedolizumab and TNF antagonists showed comparable risks of SIs in other studies besides this one by Kirchgesner et al. (2022), Singh et al. (2022). In addition, in the study by Bressler B et al., Vedolizumab-treated patients were less likely to experience SAEs compared with TNF antagonists after adjustment using inverse probability weighting (Bressler et al., 2021). To clarify this ambiguity, head-to-head, direct comparisons and more real-world studies were needed. Second, the treatment of CD is a lifelong process, clearly, even the RCT with the longest study duration of 60 weeks included in our analysis, would not be able to cover the entire course of a CD patient, and therefore interpretations of these safety findings are restricted. In several long-term open-label studies, multiple agents, included Ustekinumab and Vedolizumab, maintained good efficacy in maintenance of remission and no new safety signal was observed, although LOR continues to be a problem in later treatment (Vermeire et al., 2017; Loftus et al., 2020; Sandborn et al., 2022). Third, safety information of medical products from RCTs designed for registrational purposes is often underdetected and under-utilized. Indeed, the information provided in most published RCT reports is generally insufficient and inconsistent to comprehensively summarize the safety of medical products, even in RCTs reports published in high-impact journals. No significant differences in AEs rates between the interventions and placebo reported in some studies do not imply that the studying drugs are truly safe (Phillips et al., 2019). Therefore, the evaluations of safety should integrate the reported RCTs studies and real-world evidences. Consistency of reporting important safety results across trials should be improved to facilitate comprehensive comparisons, which can be referred to the work that the COMET (Core Outcome Measures in Effectiveness Trials) group contributes to (Gargon, 2016), and reporting safety data according to the CONSORT (Consolidated Standards of Reporting Trials) harm checklist should be promoted in RCTs reports (Cuschieri, 2019). Nevertheless, the dominances of Risankizumab, BI 695501 and Vedolizumab SC in risks of SAEs in induction therapy, in risks of SAEs, and of SIs in maintenance therapy, respectively, in our analysis deserve further investigation.

Our analysis has several distinct strengths over other previous meta-analyses comparing the safety and efficacy of biologics for moderate-to-severe CD. First, we utilized Bayesian methods instead of a classical Frequentist method, and employed MCMC which can leverage prior information and circumvent the limitations of hypothesis testing, for parameter estimation and sampling from the posterior distribution, and provided complete assessments of model fit and convergence of MCMC model. Second, we evaluated efficacy separately with different definitions of clinical response (CDAI-70 and CDAI-100) to reduce heterogeneity. Third, we review some of the published key results of the SEQUENCE trial, and included a direct head-to-head comparison (SEAVUE) between Ustekinumab and Adalimumab, which reduces uncertainty of the results, validates the results of indirect comparison and strengthens the comparative networks. Fourthly, we also included comparisons between different dose groups in two three-arm trials (ACCENT-1 and FORTIFY) in evaluations of maintenance therapy, which further enriches our comparative networks, and an agent, Vedolizumab SC, with potential advantages on safety and patient compliance.

Of course, we also acknowledge that there are several limitations and shortcomings in our analysis. First, we evaluated the safety and efficacy of biologics in stratified analyses according to their history of TNF antagonist treatment in CD patients, as well as in the overall CD patients. Although it might seem unnecessary, we believe this approach could be beneficial for managing patients with an ambiguous history of TNF antagonist treatment. Second, we used clinical response and clinical remission as primary efficacy outcomes for induction and maintenance therapy, respectively. Actually, in the CORE-IBD consensus initiative (first international consensus-based Core Outcome Set [COS] for use in IBD RCTs) (Collaborators et al., 2022), there was controversy among panelists about historically defining clinical remission as a CDAI score of <150. Furthermore, endoscopic outcomes, defined by the more recognized SES-CD (Simple Endoscopic Score for Crohn's Disease) (Daperno et al., 2004), and corticosteroid-free remission, which is more clinically significant and improves the sensitivity of detecting efficacy, are included in the configuration of outcomes in the CORE-IBD consensus. Although these outcomes are highly regarded, we believe that optimizing the setting of outcomes remains challenging in the face of heterogeneity in outcome definitions across RCTs distributed over several decades, as well as the lack of relevant data of outcomes and few stratified analyses according to history of TNF antagonist treatment in many RCTs. Third, we acknowledge the heterogeneity in RCTs design and baseline characteristics of studying patients between the included studies. The baseline characteristics of enrolled patients in recently completed RCTs, influenced by decades of development in biologics for treating CD, differ from those in earlier studies: the proportion of patients enrolled with a history of surgery for CD in recent RCTs seems to be lower. For example, in the VISIBLE-2 trial (2022), 26.9% (110/409) of enrolled patients with a history of surgery for CD, compared to 41.8% (466/1115) in the GEMINI-2 trial (2013), indicating a significant difference. Additionally, patients enrolled in recent RCTs had a longer duration of CD and an increased proportion with a history of biologic treatment. To control the overall heterogeneity of the pooled results, we conservatively used global random effects model, included both re-randomization and treat-though trials in our analysis, and finally assessed the impact of model selection on the robustness of the results by sensitivity analysis.

In conclusion, this updated systematic review and network meta-analysis, utilizing a Bayesian consistency random-effects model, highlights several key findings for the treatment of CD: Infliximab combined with azathioprine is particularly superior for induction therapy in TNF antagonist-naïve patients. Adalimumab and the 10 mg/kg maintenance regimen of Infliximab have maintenance efficacy benefits. Ustekinumab and Risankizumab may be preferred biologic agents for induction therapy in TNF antagonist-experienced patients. Risankizumab maybe have the advantage worth exploring of lower safety risks. In contrast, there is insufficient evidence from RCTs to support the safety advantage of Vedolizumab. Our analysis provides valuable insights for the risk-benefit assessments of biologics in CD treatment by patients and healthcare professionals.

Data availability statement

The original contributions presented in the study are included in the article/Supplementary Material, further inquiries can be directed to the corresponding authors.

Author contributions

HS: curation, Formal Analysis, Investigation, Data Methodology, Software, Visualization, Writing-original draft. SX: Conceptualization, Data curation, Investigation, Writing-review and editing. ZL: Visualization, Software, Writing-review and editing. TX: Validation, Writing-review and editing. JZ: XY: Conceptualization, Writing-review and editing. Conceptualization, Project administration, Supervision, Writing-review and editing.

Funding

The author(s) declare that no financial support was received for the research, authorship, and/or publication of this article.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fphar.2024.1475222/ full#supplementary-material

References

Alric, H., Amiot, A., Kirchgesner, J., Treton, X., Allez, M., Bouhnik, Y., et al. (2020). The effectiveness of either ustekinumab or vedolizumab in 239 patients with Crohn's disease refractory to anti-tumour necrosis factor. *Aliment. Pharmacol. Ther.* 51 (10), 948–957. doi:10.1111/apt.15706

Beilman, C. L., Kirwin, E., Ma, C., McCabe, C., Fedorak, R. N., and Halloran, B. (2019). Early initiation of tumor necrosis factor antagonist-based therapy for patients with crohn's disease reduces costs compared with late initiation. *Clin. Gastroenterol. Hepatol.* 17 (8), 1515–1524. doi:10.1016/j.cgh.2018.07.032

Beliveau, A., Boyne, D. J., Slater, J., Brenner, D., and Arora, P. (2019). BUGSnet: an R package to facilitate the conduct and reporting of Bayesian network Meta-analyses. BMC Med. Res. Methodol. 19 (1), 196. doi:10.1186/s12874-019-0829-2

Ben-Horin, S., and Chowers, Y. (2011). Review article: loss of response to anti-TNF treatments in Crohn's disease. *Aliment. Pharmacol. Ther.* 33 (9), 987–995. doi:10.1111/j. 1365-2036.2011.04612.x

Benson, J. M., Peritt, D., Scallon, B. J., Heavner, G. A., Shealy, D. J., Giles-Komar, J. M., et al. (2011). Discovery and mechanism of ustekinumab: a human monoclonal antibody targeting interleukin-12 and interleukin-23 for treatment of immune-mediated disorders. *MAbs* 3 (6), 535–545. doi:10.4161/mabs.3.6.17815

Biemans, V. B. C., van der Woude, C. J., Dijkstra, G., van der Meulen-de Jong, A. E., Lowenberg, M., de Boer, N. K., et al. (2020). Ustekinumab is associated with superior effectiveness outcomes compared to vedolizumab in Crohn's disease patients with prior failure to anti-TNF treatment. *Aliment. Pharmacol. Ther.* 52 (1), 123–134. doi:10.1111/apt. 15745

Borenstein, M., Hedges, L. V., Higgins, J. P., and Rothstein, H. R. (2010). A basic introduction to fixed-effect and random-effects models for meta-analysis. *Res. Synth. Methods* 1 (2), 97–111. doi:10.1002/jrsm.12

Bouhnik, Y., Carbonnel, F., Laharie, D., Stefanescu, C., Hébuterne, X., Abitbol, V., et al. (2018). Efficacy of adalimumab in patients with Crohn's disease and symptomatic small bowel stricture: a multicentre, prospective, observational cohort (CREOLE) study. *Gut* 67 (1), 53–60. doi:10.1136/gutjnl-2016-312581

Bressler, B., Yarur, A., Silverberg, M. S., Bassel, M., Bellaguarda, E., Fourment, C., et al. (2021). Vedolizumab and anti-tumour necrosis factor α real-world outcomes in biologic-naïve inflammatory bowel disease patients: results from the EVOLVE study. *J. Crohns Colitis* 15 (10), 1694–1706. doi:10.1093/ecco-jcc/jjab058

Brunda, M. J., Luistro, L., Warrier, R. R., Wright, R. B., Hubbard, B. R., Murphy, M., et al. (1993). Antitumor and antimetastatic activity of interleukin 12 against murine tumors. *J. Exp. Med.* 178 (4), 1223–1230. doi:10.1084/jem.178.4.1223

Chen, B., Gao, X., Zhong, J., Ren, J., Zhu, X., Liu, Z., et al. (2020). Efficacy and safety of adalimumab in Chinese patients with moderately to severely active Crohn's disease: results from a randomized trial. *Ther. Adv. Gastroenterol.* 13, 1756284820938960. doi:10.1177/1756284820938960

Collaborators, C.-I., Ma, C., Hanzel, J., Panaccione, R., Sandborn, W. J., D'Haens, G. R., et al. (2022). CORE-IBD: a multidisciplinary international consensus initiative to develop a Core outcome set for randomized controlled trials in inflammatory bowel disease. *Gastroenterology* 163 (4), 950–964. doi:10.1053/j. gastro.2022.06.068

Colombel, J. F., Hanauer, S. B., Sandborn, W., Sands, B. E., Schreiber, S., Danese, S., et al. (2023). DOP86 Subcutaneous infliximab (CT-P13 SC) as maintenance therapy for Crohn's disease: a phase 3, randomised, placebo-controlled study (LIBERTY-CD). J. Crohn's Colitis 17 (Suppl. ment_1), i161-i162. doi:10.1093/ecco-jcc/jjac190. 0126

Colombel, J. F., Sandborn, W. J., Reinisch, W., Mantzaris, G. J., Kornbluth, A., Rachmilewitz, D., et al. (2010). Infliximab, azathioprine, or combination therapy for Crohn's disease. *N. Engl. J. Med.* 362 (15), 1383–1395. doi:10.1056/NEJMoa0904492

Colombel, J. F., Sandborn, W. J., Rutgeerts, P., Enns, R., Hanauer, S. B., Panaccione, R., et al. (2007). Adalimumab for maintenance of clinical response and remission in patients with Crohn's disease: the CHARM trial. *Gastroenterology* 132 (1), 52–65. doi:10.1053/j.gastro.2006.11.041

Cuschieri, S. (2019). The CONSORT statement. Saudi J. Anaesth. 13 (Suppl. 1), S27–S30. doi:10.4103/sja.SJA_559_18

Daperno, M., D'Haens, G., Van Assche, G., Baert, F., Bulois, P., Maunoury, V., et al. (2004). Development and validation of a new, simplified endoscopic activity score for Crohn's disease: the SES-CD. *Gastrointest. Endosc.* 60 (4), 505–512. doi:10.1016/s0016-5107(04)01878-4

D'Haens, G., Panaccione, R., Baert, F., Bossuyt, P., Colombel, J. F., Danese, S., et al. (2022). Risankizumab as induction therapy for Crohn's disease: results from the phase 3 ADVANCE and MOTIVATE induction trials. *Lancet* 399 (10340), 2015–2030. doi:10.1016/S0140-6736(22)00467-6

Dias, S., Welton, N. J., Caldwell, D. M., and Ades, A. E. (2010). Checking consistency in mixed treatment comparison meta-analysis. *Stat. Med.* 29 (7-8), 932–944. doi:10. 1002/sim.3767

Dias, S., Welton, N. J., Sutton, A. J., and Ades, A. E. (2014a). in *NICE dsu technical* support document 2: a generalised linear modelling framework for pairwise and network meta-analysis of randomised controlled trials. London.

Dias, S., Welton, N. J., Sutton, A. J., Caldwell, D. M., Lu, G., and Ades, A. E. (2014b). in NICE DSU technical support document 4: inconsistency in networks of evidence based on randomised controlled trials. London.

Feagan, B. G., Sandborn, W. J., D'Haens, G., Panes, J., Kaser, A., Ferrante, M., et al. (2017). Induction therapy with the selective interleukin-23 inhibitor risankizumab in patients with moderate-to-severe Crohn's disease: a randomised, double-blind, placebo-controlled phase 2 study. *Lancet* 389 (10080), 1699–1709. doi:10.1016/S0140-6736(17) 30570-6

Feagan, B. G., Sandborn, W. J., Gasink, C., Jacobstein, D., Lang, Y., Friedman, J. R., et al. (2016). Ustekinumab as induction and maintenance therapy for crohn's disease. *N. Engl. J. Med.* 375 (20), 1946–1960. doi:10.1056/NEJMoa1602773

Ferrante, M., Panaccione, R., Baert, F., Bossuyt, P., Colombel, J. F., Danese, S., et al. (2022). Risankizumab as maintenance therapy for moderately to severely active Crohn's disease: results from the multicentre, randomised, double-blind, placebo-controlled, withdrawal phase 3 FORTIFY maintenance trial. *Lancet* 399 (10340), 2031–2046. doi:10.1016/S0140-6736(22)00466-4

Garcia, M. J., Rivero, M., Fernandez-Clotet, A., de Francisco, R., Sicilia, B., Mesonero, F., et al. (2024). Comparative study of the effectiveness of vedolizumab versus ustekinumab after anti-TNF failure in crohn's disease (Versus-CD): data from the ENEIDA registry. J. Crohns Colitis 18 (1), 65–74. doi:10.1093/ecco-jcc/jjad124

Gargon, E. (2016). The COMET (Core outcome Measures in effectiveness trials) initiative. *Maturitas* 91, 91–92. doi:10.1016/j.maturitas.2016.06.007

Hanauer, S., Liedert, B., Balser, S., Brockstedt, E., Moschetti, V., and Schreiber, S. (2021). Safety and efficacy of BI 695501 versus adalimumab reference product in patients with advanced Crohn's disease (VOLTAIRE-CD): a multicentre, randomised, double-blind, phase 3 trial. *Lancet Gastroenterol. Hepatol.* 6 (10), 816–825. doi:10.1016/S2468-1253(21)00252-1

Hanauer, S. B., Feagan, B. G., Lichtenstein, G. R., Mayer, L. F., Schreiber, S., Colombel, J. F., et al. (2002). Maintenance infliximab for Crohn's disease: the ACCENT I randomised trial. *Lancet* 359 (9317), 1541–1549. doi:10.1016/S0140-6736(02)08512-4

Hanauer, S. B., Sandborn, W. J., Rutgeerts, P., Fedorak, R. N., Lukas, M., MacIntosh, D., et al. (2006). Human anti-tumor necrosis factor monoclonal antibody (adalimumab) in Crohn's disease: the CLASSIC-I trial. *Gastroenterology* 130 (2), 323–333. quiz 591. doi:10.1053/j.gastro.2005.11.030

Higgins, J. P. T., and Cochrane, C. (2019). Cochrane handbook for systematic reviews of interventions. London: Cochrane.

Hisamatsu, T., Kato, S., Kunisaki, R., Matsuura, M., Nagahori, M., Motoya, S., et al. (2019). Withdrawal of thiopurines in Crohn's disease treated with scheduled adalimumab maintenance: a prospective randomised clinical trial (DIAMOND2). *J. Gastroenterol.* 54 (10), 860–870. doi:10.1007/s00535-019-01582-w

Hoaglin, D. C., Hawkins, N., Jansen, J. P., Scott, D. A., Itzler, R., Cappelleri, J. C., et al. (2011). Conducting indirect-treatment-comparison and network-meta-analysis studies: report of the ISPOR task force on indirect treatment comparisons good research practices: part 2. *Value Health* 14 (4), 429–437. doi:10.1016/j.jval.2011.01.011

Hu, A., Kotze, P. G., Burgevin, A., Tan, W., Jess, A., Li, P. S., et al. (2021). Combination therapy does not improve rate of clinical or endoscopic remission in patients with inflammatory bowel diseases treated with vedolizumab or ustekinumab. *Clin. Gastroenterol. Hepatol.* 19 (7), 1366–1376.e2. doi:10.1016/j.cgh.2020.07.012

Hutton, B., Salanti, G., Caldwell, D. M., Chaimani, A., Schmid, C. H., Cameron, C., et al. (2015). The PRISMA extension statement for reporting of systematic reviews incorporating network meta-analyses of health care interventions: checklist and explanations. *Ann. Intern Med.* 162 (11), 777–784. doi:10.7326/M14-2385

Kanters, S. (2022). Fixed- and random-effects models. *Methods Mol. Biol.* 2345, 41-65. doi:10.1007/978-1-0716-1566-9_3

Kirchgesner, J., Desai, R. J., Beaugerie, L., Schneeweiss, S., and Kim, S. C. (2022). Risk of serious infections with vedolizumab versus tumor necrosis factor antagonists in patients with inflammatory bowel disease. *Clin. Gastroenterol. Hepatol.* 20 (2), 314–324.e16. doi:10.1016/j.cgh.2020.12.030

Knight, D. M., Trinh, H., Le, J., Siegel, S., Shealy, D., McDonough, M., et al. (1993). Construction and initial characterization of a mouse-human chimeric anti-TNF antibody. *Mol. Immunol.* 30 (16), 1443–1453. doi:10.1016/0161-5890(93)90106-1

Lee, M. J., Parker, C. E., Taylor, S. R., Guizzetti, L., Feagan, B. G., Lobo, A. J., et al. (2018). Efficacy of medical therapies for fistulizing crohn's disease: systematic review and meta-analysis. *Clin. Gastroenterol. Hepatol.* 16 (12), 1879–1892. doi:10.1016/j.cgh. 2018.01.030

Loftus, E. V., Jr., Feagan, B. G., Panaccione, R., Colombel, J. F., Sandborn, W. J., Sands, B. E., et al. (2020). Long-term safety of vedolizumab for inflammatory bowel disease. *Aliment. Pharmacol. Ther.* 52 (8), 1353–1365. doi:10.1111/apt.16060

Manlay, L., Boschetti, G., Pereira, B., Flourie, B., Dapoigny, M., Reymond, M., et al. (2021). Comparison of short- and long-term effectiveness between ustekinumab and vedolizumab in patients with Crohn's disease refractory to anti-tumour necrosis factor therapy. *Aliment. Pharmacol. Ther.* 53 (12), 1289–1299. doi:10.1111/apt.16377

Marchetti, M., Liberato, N. L., Di Sabatino, A., and Corazza, G. R. (2013). Costeffectiveness analysis of top-down versus step-up strategies in patients with newly diagnosed active luminal Crohn's disease. *Eur. J. Health Econ.* 14 (6), 853–861. doi:10. 1007/s10198-012-0430-7

Matsumoto, T., Motoya, S., Watanabe, K., Hisamatsu, T., Nakase, H., Yoshimura, N., et al. (2016). Adalimumab monotherapy and a combination with azathioprine for crohn's disease: a prospective, randomized trial. *J. Crohns Colitis* 10 (11), 1259–1266. doi:10.1093/ecco-jcc/jjw152

Nct (2020). Study comparing intravenous (IV)/Subcutaneous (SC) risankizumab to IV/SC ustekinumab to assess change in crohn's disease activity index (CDAI) in adult participants with moderate to severe crohn's disease (CD). Available at: https:// clinicaltrials.gov/show/NCT04524611.

Phillips, R., Hazell, L., Sauzet, O., and Cornelius, V. (2019). Analysis and reporting of adverse events in randomised controlled trials: a review. *BMJ Open* 9 (2), e024537. doi:10.1136/bmjopen-2018-024537

Present, D. H., Rutgeerts, P., Targan, S., Hanauer, S. B., Mayer, L., van Hogezand, R. A., et al. (1999). Infliximab for the treatment of fistulas in patients with Crohn's disease. *N. Engl. J. Med.* 340 (18), 1398–1405. doi:10.1056/nejm199905063401804

Puhan, M. A., Schunemann, H. J., Murad, M. H., Li, T., Brignardello-Petersen, R., Singh, J. A., et al. (2014). A GRADE Working Group approach for rating the quality of treatment effect estimates from network meta-analysis. *BMJ* 349, g5630. doi:10.1136/ bmj.g5630

Rodríguez-Lago, I., Hoyo, J. D., Pérez-Girbés, A., Garrido-Marín, A., Casanova, M. J., Chaparro, M., et al. (2020). Early treatment with anti-tumor necrosis factor agents improves long-term effectiveness in symptomatic stricturing Crohn's disease. *United Eur. Gastroenterol. J.* 8 (9), 1056–1066. doi:10.1177/2050640620947579

Rutgeerts, P., D'Haens, G., Targan, S., Vasiliauskas, E., Hanauer, S. B., Present, D. H., et al. (1999). Efficacy and safety of retreatment with anti-tumor necrosis factor antibody (infliximab) to maintain remission in Crohn's disease. *Gastroenterology* 117 (4), 761–769. doi:10.1016/s0016-5085(99)70332-x

Rutgeerts, P., Van Assche, G., Sandborn, W. J., Wolf, D. C., Geboes, K., Colombel, J. F., et al. (2012). Adalimumab induces and maintains mucosal healing in patients with Crohn's disease: data from the EXTEND trial. *Gastroenterology* 142 (5), 1102–1111.e2. doi:10.1053/j.gastro.2012.01.035

Sandborn, W. J., Feagan, B. G., Rutgeerts, P., Hanauer, S., Colombel, J. F., Sands, B. E., et al. (2013). Vedolizumab as induction and maintenance therapy for Crohn's disease. *N. Engl. J. Med.* 369 (8), 711–721. doi:10.1056/NEJMoa1215739

Sandborn, W. J., Feagan, B. G., Stoinov, S., Honiball, P. J., Rutgeerts, P., Mason, D., et al. (2007a). Certolizumab pegol for the treatment of Crohn's disease. *N. Engl. J. Med.* 357 (3), 228–238. doi:10.1056/NEJMoa067594

Sandborn, W. J., Gasink, C., Gao, L. L., Blank, M. A., Johanns, J., Guzzo, C., et al. (2012). Ustekinumab induction and maintenance therapy in refractory Crohn's disease. *N. Engl. J. Med.* 367 (16), 1519–1528. doi:10.1056/NEIMoa1203572

Sandborn, W. J., Hanauer, S. B., Rutgeerts, P., Fedorak, R. N., Lukas, M., MacIntosh, D. G., et al. (2007b). Adalimumab for maintenance treatment of Crohn's disease: results of the CLASSIC II trial. *Gut* 56 (9), 1232–1239. doi:10.1136/gut.2006.106781

Sandborn, W. J., Rebuck, R., Wang, Y., Zou, B., Adedokun, O. J., Gasink, C., et al. (2022). Five-Year efficacy and safety of ustekinumab treatment in crohn's disease: the IM-UNITI trial. *Clin. Gastroenterol. Hepatol.* 20 (3), 578–590.e4. doi:10.1016/j.cgh. 2021.02.025

Sandborn, W. J., Rutgeerts, P., Enns, R., Hanauer, S. B., Colombel, J. F., Panaccione, R., et al. (2007c). Adalimumab induction therapy for Crohn disease previously treated with infliximab: a randomized trial. *Ann. Intern Med.* 146 (12), 829–838. doi:10.7326/0003-4819-146-12-200706190-00159

Sandborn, W. J., Schreiber, S., Feagan, B. G., Rutgeerts, P., Younes, Z. H., Bloomfield, R., et al. (2011). Certolizumab pegol for active Crohn's disease: a placebo-controlled, randomized trial. *Clin. Gastroenterol. Hepatol.* 9 (8), 670–678. doi:10.1016/j.cgh.2011. 04.031

Sands, B. E., Anderson, F. H., Bernstein, C. N., Chey, W. Y., Feagan, B. G., Fedorak, R. N., et al. (2004). Infliximab maintenance therapy for fistulizing Crohn's disease. *N. Engl. J. Med.* 350 (9), 876–885. doi:10.1056/NEJMoa030815

Sands, B. E., Feagan, B. G., Rutgeerts, P., Colombel, J. F., Sandborn, W. J., Sy, R., et al. (2014). Effects of vedolizumab induction therapy for patients with Crohn's disease in whom tumor necrosis factor antagonist treatment failed. *Gastroenterology* 147 (3), 618–627. doi:10.1053/j.gastro.2014.05.008

Sands, B. E., Irving, P. M., Hoops, T., Izanec, J. L., Gao, L. L., Gasink, C., et al. (2022). Ustekinumab versus adalimumab for induction and maintenance therapy in biologicnaive patients with moderately to severely active Crohn's disease: a multicentre, randomised, double-blind, parallel-group, phase 3b trial. *Lancet* 399 (10342), 2200–2211. doi:10.1016/S0140-6736(22)00688-2

Sands, B. E., Steinhart, A. H., Lewis, J. D., Hanauer, S. B., Persson, T., and Sandborn, W. J. (2024). Optimal Crohn's disease activity index (CDAI) response criteria is defined by decreaseÅ100 points. *Gastroenterology* 124 (4), A206. doi:10.1016/s0016-5085(03) 81034-x

Schmitt, H., Billmeier, U., Dieterich, W., Rath, T., Sonnewald, S., Reid, S., et al. (2019). Expansion of IL-23 receptor bearing TNFR2+ T cells is associated with molecular resistance to anti-TNF therapy in Crohn's disease. *Gut* 68 (5), 814–828. doi:10.1136/ gutjnl-2017-315671 Schreiber, S., Ben-Horin, S., Leszczyszyn, J., Dudkowiak, R., Lahat, A., Gawdis-Wojnarska, B., et al. (2021). Randomized controlled trial: subcutaneous vs intravenous infliximab CT-P13 maintenance in inflammatory bowel disease. *Gastroenterology* 160 (7), 2340–2353. doi:10.1053/j.gastro.2021.02.068

Schreiber, S., Khaliq-Kareemi, M., Lawrance, I. C., Thomsen, O. O., Hanauer, S. B., McColm, J., et al. (2007). Maintenance therapy with certolizumab pegol for Crohn's disease. *N. Engl. J. Med.* 357 (3), 239–250. doi:10.1056/NEJMoa062897

Singh, S., Heien, H. C., Herrin, J., Dulai, P. S., Sangaralingham, L., Shah, N. D., et al. (2022). Comparative risk of serious infections with tumor necrosis factor α antagonists vs vedolizumab in patients with inflammatory bowel diseases. *Clin. Gastroenterol. Hepatol.* 20 (2), e74–e88. doi:10.1016/j.cgh.2021.02.032

Singh, S., Heien, H. C., Sangaralingham, L. R., Schilz, S. R., Kappelman, M. D., Shah, N. D., et al. (2016). Comparative effectiveness and safety of anti-tumor necrosis factor agents in biologic-naive patients with crohn's disease. *Clin. Gastroenterol. Hepatol.* 14 (8), 1120–1129. doi:10.1016/j.cgh.2016.03.038

Singh, S., Kim, J., Luo, J., Paul, P., Rudrapatna, V., Park, S., et al. (2023). Comparative safety and effectiveness of biologic therapy for crohn's disease: a CA-IBD cohort study. *Clin. Gastroenterol. Hepatol.* 21 (9), 2359–2369.e5. doi:10.1016/j.cgh.2022.10.029

Singh, S., Kroe-Barrett, R. R., Canada, K. A., Zhu, X., Sepulveda, E., Wu, H., et al. (2015). Selective targeting of the IL23 pathway: generation and characterization of a novel high-affinity humanized anti-IL23A antibody. *MAbs* 7 (4), 778–791. doi:10.1080/19420862.2015.1032491

Smith, P. J., Critchley, L., Storey, D., Gregg, B., Stenson, J., Kneebone, A., et al. (2022). Efficacy and safety of elective switching from intravenous to subcutaneous infliximab [CT-P13]: a multicentre cohort study. *J. Crohns Colitis* 16 (9), 1436–1446. doi:10.1093/ ecco-jcc/jjac053

Soler, D., Chapman, T., Yang, L. L., Wyant, T., Egan, R., and Fedyk, E. R. (2009). The binding specificity and selective antagonism of vedolizumab, an anti-alpha4beta7 integrin therapeutic antibody in development for inflammatory bowel diseases. *J. Pharmacol. Exp. Ther.* 330 (3), 864–875. doi:10.1124/jpet.109.153973

Sterne, J. A. C., Savovic, J., Page, M. J., Elbers, R. G., Blencowe, N. S., Boutron, I., et al. (2019). RoB 2: a revised tool for assessing risk of bias in randomised trials. *BMJ* 366, 14898. doi:10.1136/bmj.14898

Stobie, L., Gurunathan, S., Prussin, C., Sacks, D. L., Glaichenhaus, N., Wu, C. Y., et al. (2000). The role of antigen and IL-12 in sustaining Th1 memory cells *in vivo*: IL-12 is required to maintain memory/effector Th1 cells sufficient to mediate protection to an infectious parasite challenge. *Proc. Natl. Acad. Sci. U. S. A.* 97 (15), 8427–8432. doi:10. 1073/pnas.160197797

Sutton, A. J., and Abrams, K. R. (2001). Bayesian methods in meta-analysis and evidence synthesis. *Stat. Methods Med. Res.* 10 (4), 277–303. doi:10.1177/096228020101000404

Takeda (2017). Vedolizumab intravenous (IV) compared to placebo in Chinese participants with crohn's disease. Available at: https://classic.clinicaltrials.gov/show/ NCT03234907.

Targan, S. R., Hanauer, S. B., van Deventer, S. J., Mayer, L., Present, D. H., Braakman, T., et al. (1997). A short-term study of chimeric monoclonal antibody cA2 to tumor necrosis factor alpha for Crohn's disease. Crohn's Disease cA2 Study Group. *N. Engl. J. Med.* 337 (15), 1029–1035. doi:10.1056/NEJM199710093371502

Thompson, W., and Argaez, C. (2019). in Early biologic treatment versus conventional treatment for the management of crohn's disease: a review of comparative clinical effectiveness and cost-effectiveness. Ottawa (ON).

Townsend, T., Razanskaite, V., Dodd, S., Storey, D., Michail, S., Morgan, J., et al. (2020). Comparative effectiveness of ustekinumab or vedolizumab after one year in 130 patients with anti-TNF-refractory Crohn's disease. *Aliment. Pharmacol. Ther.* 52 (8), 1341–1352. doi:10.1111/apt.16057

Ungaro, R. C., Aggarwal, S., Topaloglu, O., Lee, W. J., Clark, R., and Colombel, J. F. (2020). Systematic review and meta-analysis: efficacy and safety of early biologic treatment in adult and paediatric patients with Crohn's disease. *Aliment. Pharmacol. Ther.* 51 (9), 831–842. doi:10.1111/apt.15685

Van Assche, G., Magdelaine-Beuzelin, C., D'Haens, G., Baert, F., Noman, M., Vermeire, S., et al. (2008). Withdrawal of immunosuppression in Crohn's disease treated with scheduled infliximab maintenance: a randomized trial. *Gastroenterology* 134 (7), 1861–1868. doi:10.1053/j.gastro.2008.03.004

van Dullemen, H. M., van Deventer, S. J., Hommes, D. W., Bijl, H. A., Jansen, J., Tytgat, G. N., et al. (1995). Treatment of Crohn's disease with anti-tumor necrosis factor chimeric monoclonal antibody (cA2). *Gastroenterology* 109 (1), 129–135. doi:10.1016/ 0016-5085(95)90277-5

van Valkenhoef, G., Lu, G., de Brock, B., Hillege, H., Ades, A. E., and Welton, N. J. (2012). Automating network meta-analysis. *Res. Synth. Methods* 3 (4), 285–299. doi:10. 1002/jrsm.1054

Vavricka, S. R., Gubler, M., Gantenbein, C., Spoerri, M., Froehlich, F., Seibold, F., et al. (2017). Anti-TNF treatment for extraintestinal manifestations of inflammatory bowel disease in the Swiss IBD cohort study. *Inflamm. Bowel Dis.* 23 (7), 1174–1181. doi:10. 1097/mib.00000000001109

Vermeire, S., D'Haens, G., Baert, F., Danese, S., Kobayashi, T., Loftus, E. V., et al. (2022). Efficacy and safety of subcutaneous vedolizumab in patients with moderately to

severely active crohn's disease: results from the VISIBLE 2 randomised trial. J. Crohns Colitis 16 (1), 27–38. doi:10.1093/ecco-jcc/jjab133

Vermeire, S., Loftus, E. V., Jr., Colombel, J. F., Feagan, B. G., Sandborn, W. J., Sands, B. E., et al. (2017). Long-term efficacy of vedolizumab for crohn's disease. *J. Crohns Colitis* 11 (4), 412–424. doi:10.1093/ecco-jcc/jjw176

Watanabe, K., Motoya, S., Ogata, H., Kanai, T., Matsui, T., Suzuki, Y., et al. (2020). Effects of vedolizumab in Japanese patients with Crohn's disease: a prospective, multicenter, randomized, placebo-controlled Phase 3 trial with exploratory analyses. *J. Gastroenterol.* 55 (3), 291–306. doi:10.1007/s00535-019-01647-w

Watanabe, M., Hibi, T., Lomax, K. G., Paulson, S. K., Chao, J., Alam, M. S., et al. (2012). Adalimumab for the induction and maintenance of clinical remission in Japanese patients with Crohn's disease. *J. Crohns Colitis* 6 (2), 160–173. doi:10.1016/j.crohns.2011.07.013

Watt, J., and Del Giovane, C. (2022). Network meta-analysis. *Methods Mol. Biol.* 2345, 187–201. doi:10.1007/978-1-0716-1566-9_12

Wong, U., and Cross, R. K. (2019). Expert opinion on interleukin-12/23 and interleukin-23 antagonists as potential therapeutic options for the treatment of inflammatory bowel disease. *Expert Opin. Investig. Drugs* 28 (5), 473–479. doi:10. 1080/13543784.2019.1597053

Wyant, T., Fedyk, E., and Abhyankar, B. (2016). An overview of the mechanism of action of the monoclonal antibody vedolizumab. *J. Crohns Colitis* 10 (12), 1437–1444. doi:10.1093/ecco-jcc/jjw092

Ye, B. D., Pesegova, M., Alexeeva, O., Osipenko, M., Lahat, A., Dorofeyev, A., et al. (2019). Efficacy and safety of biosimilar CT-P13 compared with originator infliximab in patients with active Crohn's disease: an international, randomised, double-blind, phase 3 non-inferiority study. *Lancet* 393 (10182), 1699–1707. doi:10.1016/S0140-6736(18) 32196-2

Yzet, C., Diouf, M., Singh, S., Brazier, F., Turpin, J., Nguyen-Khac, E., et al. (2021). No benefit of concomitant immunomodulator therapy on efficacy of biologics that are not tumor necrosis factor antagonists in patients with inflammatory bowel diseases: a metaanalysis. *Clin. Gastroenterol. Hepatol.* 19 (4), 668–679.e8. doi:10.1016/j.cgh.2020.06.071